Author’s response to reviews

Title: Testing the impact of a social skill training versus waiting list control group for the reduction of disruptive behaviors and stress among preschool children in child care: The study protocol for a cluster randomized trial

Authors:

Sylvana Côté (sylvana.cote.1@umontreal.ca)

Marie-Pier Larose (marie.pier.larose@gmail.com)

Marie-Claude Geoffroy (marie-claude.geoffroy@mcgill.ca)

Julie Laurin (j.laurin@umontreal.ca)

Frank Vitaro (frank.vitaro@umontreal.ca)

Richard Tremblay (tremlar@grip.umontreal.ca)

Isabelle Ouellet-Morin (isabelle.ouellet-morin@umontreal.ca)

Version: 2 Date: 21 Jul 2017

Author’s response to reviews:

1. We add the mention "ongoing study" at the end of the abstract of well as in the Method' section of the manuscript.

2. Authorship has been revised.